Author's response to reviews

Title: Drowning in Aboriginal and Torres Strait Islander Children and Adolescents in Queensland (Australia)

Authors:

Belinda A Wallis Ms (Belinda.Wallis@uqconnect.edu.au)
Kerrianne Watt A/Prof (Kerrianne.Watt@jcu.edu.au)
Richard C Franklin A/Prof (Richarad.Franklin@jcu.edu.au)
Roy M Kimble Prof (royk@uq.edu.au)

Version: 2 Date: 13 March 2015

Author's response to reviews:

Line numbers have been inserted and page breaks deleted except for the Table 3 which appears in Landscape format after a section break.
Competing interests have been inserted and are declared as none.